Author’s response to reviews

Title: Changing Behaviour 'more or less' - Do theories of behaviour inform strategies for implementation and de-implementation? A Critical Interpretive Synthesis

Authors:
Andrea Patey (apatey@ohri.ca)
Catherine Hurt (catherine.hurt.1@city.ac.uk)
Jeremy Grimshaw (jgrimshaw@ohri.ca)
Jill Francis (jill.francis.1@city.ac.uk)

Version: 1 Date: 15 May 2018

Author’s response to reviews:

See attached Letter